Author's response to reviews

Title: Effectiveness and efficiency of a practice accreditation program on cardiovascular risk management in primary care: study protocol of a clustered randomized trial.

Authors:

Elvira Nouwens (e.nouwens@iq.umcn.nl)  
Jan van Lieshout (j.vanlieshout@iq.umcn.nl)  
Eddy Adang (e.adang@ebh.umcn.nl)  
Margriet Bouma (m.bouma@nhg.org)  
Jozé Braspenning (j.braspenning@iq.umcn.nl)  
Michel Wensing (m.wensing@iq.umcn.nl)

Version: 3 Date: 27 September 2012

Author's response to reviews: see over
Dear Professor Eccles,

Herewith we submit to Implementation Science the final version of our article ‘Effectiveness and efficiency of a practice accreditation program on cardiovascular risk management in primary care: study protocol of a clustered randomized trial’.

We adjusted our manuscript according to all suggested comments. Furthermore, we added a sentence in the discussion (the last sentence) to explain no data cleaning or analysis has occurred prior to submission of the manuscript, as suggested by Robbie Foy in an e-mail we received on the 27th of August. Hopefully the manuscript is now acceptable for publication. If there are more adjustments to be made, please let me know.

On behalf of all authors,

Yours sincerely,

Elvira Nouwens, MSc
Radboud University Nijmegen Medical Centre, The Netherlands